Author's response to reviews

Title: Genetic mutations in acute pancreatitis

Authors:

Dorota Koziel (dorota.koziel@wp.pl)
Stanislaw Gluszek (sgluszek@wp.pl)
Artur Kowalik (artur_kowalik@yahoo.com)
Malgorzata Chlopek (malgo.chlopek@gmail.com)
Liliana Pieciak (lilianamazurek@o2.pl)

Version: 3 Date: 11 September 2014

Author's response to reviews:

Changes
1. There is no page break in the manuscript now.
2. The Abstract has been reformatted including your comments.
3. Introduction has been renamed to “Background”.
4. Figure legends are now at the end of the document.
5. Tables are now after references.
6. There is a list of abbreviations at the end of the manuscript.